Author's response to reviews

Title: Treatment of progressive supranuclear palsy with autologous adipose tissue-derived mesenchymal stem cells: a case report

Authors:

Soo Won Choi (love575dr@naver.com)
Kwon Byong Park (nskbpark@hanmail.net)
Sang Kyu Woo (paulskwoo@rnl.co.kr)
Sung Keun Kang (kangsk@rnl.co.kr)
Jeong Chan Ra (jcra@rnl.co.kr)

Version: 3 Date: 28 June 2013

Author's response to reviews:

To whom it may concern.

Thank you for your attention.

We revised the manuscript according your suggestion.

1. In keeping with journal style, we removed the sub-sections/sub-headings from the case presentation section.

2. We replaced dates found in the case presentation to the amount of time this occurred before the presentation of the case report.

3. We do have patient’s written consent and it is attached.

Thank you for your time consideration.

With regard

Dr. Jeong Chan Ra